Author's response to reviews

Title: Granulomatus cheilitis associated with exacerbations of Crohn’s disease: a case report

Authors:

John K Triantafillidis (jkt@panafonet.gr)
Flora Zervou - Valvi (jkt@panafonet.gr)
Emmanouel Merikas (jkt@panafonet.gr)
George Peros (jkt@panafonet.gr)
Ourania Nicolatou - Galitis (jkt@panafonet.gr)
Aristofanis Gikas (argikas@internet.gr)

Version: 3 Date: 6 December 2007

Author’s response to reviews: see over
Dear Editor,

Thank you for considering our case report for peer review process. We also thank the two reviewers for their comments and scientific criticism. The revised version of the manuscript according to the reviewers’ comments is submitted herewith. In addition, you may find below the responses to the comments in detail.

Looking forward to hearing from you.

Sincerely yours,

Prof. John K. Triantafillidis

Responses to the comments of Referee 2:

- We changed the title according to the suggestion of the reviewer.

- We replaced the word “enlargement” with “swelling” presented in abstract and text. Also we made all the changes regarding grammatical errors as suggested by the reviewer.

- Page 6, paragraph 2: We added relevant references.

- The whole manuscript was edited.